Author's response to reviews

Title: Analysis of positional candidate genes in the AAA1 susceptibility locus for abdominal aortic aneurysms on chromosome 19

Authors:

John H Lillvis (jlillvis@med.wayne.edu)
Yoshiki Kyo (kyo_yoshiki@hiro-hosp.jp)
Gerard Tromp (gctromp@geisinger.edu)
Guy M Lenk (glenk@umich.edu)
Ming Li (ming@epi.msu.edu)
Qing Lu (glu@epi.msu.edu)
Robert P Igo Jr (rpi@case.edu)
Natzi Sakalihasan (nsaka@chu.ulg.ac.be)
Robert E Ferrell (rferrell@pitt.edu)
Charles M Schworer (cmschworer@geisinger.edu)
Zoran Gatalica (ZORANGATALICA@creighton.edu)
Susan Land (sland@med.wayne.edu)
Helena Kuivaniemi (shkuivaniemi@geisinger.edu)

Version: 4 Date: 17 January 2011

Author's response to reviews: see over
Dear Editor:

Thank you for accepting our manuscript for publication in BMC Medical Genetics!

We have made the following editorial changes:

1) Additional files: all additional files are now numbered according to the order they are mentioned in the text.

2) Affiliations: We have included the country in the affiliation details.

3) Tables: we have modified the table titles to appear above the tables as one sentence long. Any other text now appears below the table.

4) We have changed the way we refer in the text to the figures or tables in the additional files: “Additional file 1, Table S1” etc.

5) We have made a few typographical changes to the text.

We appreciate the effort you have put into reviewing our manuscript to make it acceptable for BMC Medical Genetics.

Yours sincerely,

Helena Kuivaniemi, MD, PhD, FAHA